Author's response to reviews

Title: Abnormal expression and processing of the proprotein convertases PC1 and PC2 in human colorectal liver metastases.

Authors:

George N Tzimas (george_tzimas@hotmail.com)
Eric Chevet (eric.chevet@mcgill.ca)
Sarah Jenna (sarah.jenna@mcgill.ca)
Duc thang Nguyen (duc.thang.nguyen@mail.mcgill.ca)
Abdel M Khatib (mkhatib@ohri.ca)
Victoria Marcus (victoria.marcus@muhc.mcgill.ca)
Yi Zhang (yi.zhang@muhc.mcgill.ca)
Michel Chretien (mchretien@ohri.ca)
Nabil Seidah (seidah@ircm.qc.ca)
Peter Metrakos (peter.metrakos@muhc.mcgill.ca)

Version: 5 Date: 9 November 2005

Author's response to reviews:

Dear Editor,

we have removed all the supplementary files since they were provided only for review purposes. These supplementary data are not referred in the manuscript.

best regards

eric chevet